%Title: Four Pion Final States with Tagged Photons at Electron Positron Colliders
%Author: H. Czyz and J.H. Kuehn
%Published: <EM> Eur. Phys. J. </EM> <B> C 18 </B> (2001) 497-509. 
%hep-ph/0008262
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%                             4pi.latex 
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\documentclass[12pt]{article}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\pagestyle{empty}
%\renewcommand{\baselinestretch}{1.1}
\parskip4pt
\setlength{\textwidth}{16cm}
\setlength{\textheight}{22.cm}
\addtolength{\oddsidemargin}{-1.5cm}
\addtolength{\topmargin}{-1cm}
\usepackage{graphicx}
\usepackage{epsfig}
%% \input epsf.tex 
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{document}
\begin{thebibliography}{99}

\bibitem{Sakur} H.B. Thacker and J.J. Sakurai, Phys. Lett. {\bf 36B}
  (1971) 103.

\bibitem{Tsai} P. Tsai, Phys. Rev. {\bf D4}  (1971), 2821.

\bibitem{Gil} F.J. Gilman and D.H. Miller, Phys. Rev. {\bf D17} (1978) 1846;
 F.J. Gilman and S.H. Rhie Phys. Rev. {\bf D31} (1985) 1066.

\bibitem{KS} J.H.~K\"uhn, A.~Santamaria, Zeit. f. Phys.
{\bf C48} (1990) 445.

\bibitem{K1} J.H.~K\"uhn, in Proc. of the XXVIIIth RENCONTRE MORIOND,
             ed. J. Tr{\^a}n Than V{\^a}n, Editions Frontieres 1993, p.293.

\bibitem{K2} J.H.~K\"uhn, Nucl. Phys. B (Proc. Suppl.){\bf 76}(1999)21.

\bibitem{Kuraev1} H. Anlauf, A.B. Arbuzov, E.A. Kuraev, N.P. Merenkov,
 JHEP 9810:013 (1998).

\bibitem{Kuraev2} A.B. Arbuzov, E.A. Kuraev, N.P. Merenkov, L. Trentadue,
 JHEP 9812:009 (1998).

\bibitem{BKM} S.Binner, J.H. K{\"u}hn, K.Melnikov, Phys. Lett. {\bf B459}
             (1999) 279.


\bibitem{Graziano} K. Melnikov, F. Nguyen, B. Valeriani, G. Venazoni,
  Phys. Lett. {\bf B477} (2000) 114.

\bibitem{Fink} R. Decker, M. Finkemeier, P. Heiliger, H. H. Jonson,
 Z. Phys. {\bf C70} (1996) 247.

\bibitem{Pais} A. Pais, Ann. Phys.{\bf 9} (1960) 548; Ann. Phys. {\bf 22}
 (1963) 274.

\bibitem{PS} C.H. Llewellyn Smith, A. Pais, Phys. Rev. {\bf D6} (1972) 2625.

\bibitem{EI} S.I. Eidelman, V. N. Ivanchenko, 
   Nucl. Phys. B (Proc. Suppl.){\bf 76}(1999)319.

\bibitem{FWW}F. Fischer, J. Wess, F. Wagner, Z. Phys. {\bf C3} (1980) 313.

\bibitem{tauola} S. Jadach, Z. W{\c a}s, R. Decker, J. H. K{\"u}hn,
  Comp. Phys. Com. {\bf 76} (1993) 361.

\bibitem{PDG} C. Caso et al., Eur. J. Phys. {\bf C3} (1998) 1.

\bibitem{CMD2} R. R. Akhetshin et al., Phys. Lett {\bf B466} (1999) 392.

\bibitem{OLYA} L. M. Kurdadze et al., JETP Lett. {\bf 43} (1986) 643.

\bibitem{ND1} V. M. Aulchenko et al., Preprint INP 86-106, Novosibirsk,1986.

\bibitem{ND2} S. I. Dolinsky et al., Phys. Rep. {\bf 202} (1991) 99.

\bibitem{CMD} S. I. Dolinsky et al., Phys. Lett. {\bf B174} (1986) 453.

\bibitem{SND} M. N. Achasov et al., Preprint BudkerINP 98-65, Novosibirsk,1998.

\bibitem{O1} G. Cosme et al., Nucl. Phys. {\bf B152} (1979) 215.

\bibitem{O2} A. Cordier et al., Phys. Lett. {\bf 109B} (1982) 129.

\bibitem{O3} D. Bisello et al., Preprint LAL-91-64, Orsay, 1991.

\bibitem{Fr1} B. Esposito et al., Lett Nuovo Cim. {\bf 28} (1980) 195.

\bibitem{Fr2} C. Bacci et al., Phys. Lett. {\bf 95B} (1980) 139.

\bibitem{Fr3} C. Bacci et al., Nucl. Phys. {\bf B184} (1981) 31.
\bibitem{CCR}M.Caffo,H.Czy{\.z},
 E.Remiddi, Nouvo Cim. 110A(1997)515;Phys.Lett.B327(1994)369.


\end{thebibliography}

\topskip 2cm
\begin{titlepage}
%\rightline{ \large{ \bf 07 August 2000} }
\rightline{ \large{ \bf TTP00-16} }
\begin{center}
{\large\bf Four Pion Final States with Tagged Photons \\
 at Electron Positron
Colliders} \\
\end {center}
\vspace{2.cm}
\begin{center}
{\large {\bf H.~Czy{\.z}\ $^{a}$ , J.H.~K{\"u}hn$^{b}$ }
   } 

\begin{itemize}
\item[$^a$] 
             {\sl Institute of Physics, University of Silesia,
             ul. Uniwersytecka 4, 
             PL-40007 Katowice, Poland }
\item[$^b$] 
             {\sl Institut f{\"u}r Theoretische Teilchenphysik, 
              Universit{\"a}t Karlsruhe, D-76128 Karlsruhe, Germany }


\end{itemize}

\end{center}
\noindent
e-mail: {\tt  czyz@us.edu.pl\\ 
\hspace*{1.3cm} johann.kuehn@physik.uni-karlsruhe.de \\ } 
\vspace{.5cm}

\begin{center}
\begin{abstract}

A Monte Carlo generator has been constructed to
simulate the reaction $e^+e^- \to \gamma + 4 \pi$, where the photon is
assumed to be observed in the detector.
Isospin relations between the amplitudes
governing $\tau$ decays into four pions and electron positron
annihilation into four pions respectively have been found which allow to
determine all four modes after the amplitude for the $\pi^+\pi^-2\pi^0$
channel  has been fixed. The kinematic breaking of these isospin
relations as a consequence of the $\pi^-$ -- $\pi^0$ mass difference has
also been investigated.
The program is constructed in analogy to an earlier one
simulating $e^+e^- \to \gamma  + 2 \pi$. However, it does
not include final state radiation from the charged pions. Additional
collinear photon radiation has been incorporated with the technique of
structure functions. Predictions are presented for cms energies of 1GeV,
3GeV and 10GeV, corresponding to the energies of DAPHNE, BEBC and
of $B$-meson factories.
Using this program it is demonstrated that,
even after applying realistic cuts, the event rates are sufficiently high
to allow for a precise measurement of $R(Q^2)$ in the region of $Q$
between approximately 1GeV and 2.5GeV.
The model predictions are compared to recent data from electron positron
colliders and agreement is obtained within the relatively large errors.


 
\end{abstract}
\end{center}

\vfill
\end{titlepage}
\pagestyle{plain} \pagenumbering{arabic} 
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
\newcommand{\Eq}[1]{Eq.(\ref{#1})} 
%\newcommand{\labbel}[1]{\label{#1}\{#1\}} 
\newcommand{\labbel}[1]{\label{#1}} 
\newcommand{\be}{\begin{equation}}
\newcommand{\ee}{\end{equation}}
\newcommand{\ba}{\begin{eqnarray}}
\newcommand{\ea}{\end{eqnarray}}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 



\section{Introduction.}


 The precise determination of the cross section for electron positron
annihilation into hadrons over a large energy range is one of the
important tasks of current particle physics. The results are relevant
for the analysis of electroweak precision measurements which are
affected by the running of the electromagnetic coupling from the
Thompson limit up to $M_Z$. Also the interpretation of the anomalous
magnetic moment of the muon depends critically on these data. Last not
least the measurement of the energy dependence of R(s) is one of the
gold plated tests of QCD and allows for a precise determination of the
strong coupling.

Depending on the energy region different techniques for the measurement
of $R(s)$ have been applied up to date. At low energies, say from the
two pion threshold up to roughly two GeV, exclusive channels are
collected separately, for higher energies inclusive measurements start
to become dominant. For energies below $m_\tau$ isospin invariance and
CVC have traditionally been used to predict $\tau$ decays from electron
positron annihilation \cite{Sakur,Tsai,Gil,KS}. Clearly this
strategy can be inverted \cite{K1} pending irreducible uncertainties
from isospin violation and radiative corrections \cite{K2}.
At high energies a multitude of final states is present and only
inclusive measurements have been performed. 

To cover a large range of
energies, results from many different experiments and colliders have to
be combined, and energy scans have to be performed to obtain the full
energy dependence. An attractive alternative is provided by the
upcoming $\Phi-$ and $B-$ meson factories which operate at large
luminosities, albeit at fixed energies. Events with radiated tagged
photons give access to a measurement of $R$ over the full range of
energies, from threshold up to the CMS energy of the collider. For
events with tagged photons the invariant mass of the recoiling hadronic
system is fixed by the photon energy which provides an
important constraint on the event. The usage of photons  observed at
extremely  small angle with respect to the beam has been investigated in
\cite{Kuraev1,Kuraev2}. In this case final state radiation as background is
practically irrelevant. However, in practice photon detectors are not
available in this very small angular region. At large angles
  a careful study of
initial versus final state radiation has to be performed.

To arrive at reliable predictions including angular and energy cutoffs
as employed by realistic experiments, a Monte Carlo generator for the
simulation of these events is indespensible. For hadronic states with
invariant masses below two or even three GeV it is desirable to simulate
the individual exclusive channels with two, three up to six mesons i.e.
pions, kaons, ethas
 etc. which requires a fairly detailed parameterization of the
various form factors.

In principle initial and final state radiation would be required for the
complete simulation. Such a program has been constructed for the two
pion case \cite{BKM}. There it is demonstrated that suitably chosen
configurations, namely those with hard photons at small angles relative
to the beam and well separated from the pions, are dominated by initial
state radiation. In fact, this separation is possible \cite{Graziano} even
when operating the $\phi$ factory DAPHNE on top of the $\phi$ resonance
where direct radiative $\phi$ decays cannot be ignored.  

In the present paper we continue this project with the construction of a
generator for the radiative production of the four pion final state,
including the $\omega (\to3\pi) \pi$ channel. This mode contributes a
large fraction of the rate with invariant masses
  between one and two GeV. The
energy region between 1.5 GeV and 2.5 GeV is difficult to access with
current electron positron colliders. At the same time the experimental
uncertainties are relatively large. This motivates the special effort
devoted to this range.

The program is constructed in a modular form such that the
parameterization of the hadronic matrix element can easily be replaced by
a more elaborate version. Also different final states with three, four
or five pions or kaons can be included. The present
parameterization of the hadronic matrix element follows closely the form
suggested in \cite{Fink}, correcting only some minor deficiencies.
 The four pion configuration is assumed to be dominated by
 \(\rho\prime \to \pi a_1 \) plus a direct coupling 
 \(\rho\prime \to \rho\pi\pi\) and exhibits the proper behavior in the chiral
 limit.

The plan of this paper is as follows: In the next section the
formalism for the decomposition of the differential cross section into a
leptonic and a hadronic tensor is presented. Results for partially
integrated distributions are recalled which can easily be used to arrive
at simple estimates for the rates. In section 3 isospin relations are
derived between the amplitudes for four pion production from a virtual photon 
and those accessible in $\tau$ decay. The relations between these four
matrix elements contain the well known identities between the
corresponding rates but provide additional important constraints also on
the differential distributions. In section 4 the influence of the
 \(\pi^0-\pi^\pm\) mass difference on the relations obtained in section 3 
is discussed.
The ingredients of the ansatz for the
hadronic amplitude  together with a comparison between the model
prediction and the data for a variety of distributions are presented in
section 5, the complete information with all the model parameters is
collected in the appendix. A description of the Monte Carlo generator is
given in section 6, together with a few characteristic distributions. In
particular we investigate the influence of angular cuts and the addition of
collinear radiation. Section 6
contains a brief summary and the conclusions.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
\section{The radiative return.}


  Hard photons observed at small angles relative to the electron or
positron beam and at the same time well separated from charged particles
in the final state can be used to reduce the effective center of mass
energy at electron positron colliders. Performing a detailed analysis of
the angular and energy distributions for the $\gamma \pi^+ \pi^-$ final
state it has been shown that initial and final state radiation can be
reasonably well separated \cite{BKM,Graziano}. For the four pion case we
therefore restrict the discussion to initial state radiation only. The
matrix element for the production of an arbitrary hadronic final state
corresponding to the diagrams Fig.\ref{f1} are given by

 

\begin{eqnarray}
 {\cal M} = i \ e^3 \ \bar v\left(p_{+}\right) 
 \left[ \gamma^{\nu} \frac{1}{p_{-}{\kern-12pt}/ - k {\kern-5pt}/ - m}
  \epsilon^{*}{\kern-10pt}/\left(k \right)
 + \epsilon^{*}{\kern-10pt}/\left(k \right) 
    \frac{1}{k {\kern-5pt}/ - p_{+}{\kern-12pt}/ - m}\gamma^{\nu}
 \right] \frac{1}{Q^2}\ J^{em}_{\nu} \ .
\end{eqnarray}

\begin{figure}[ht]
\begin{center}
\epsfig{file=Fig1.ps,width=12cm,height=6cm}
\end{center}
\caption{Diagrams contributing to the process \(e^+e^- \to \gamma+ hadrons\)
  (only initial state radiation included).
}
\label{f1}
\end{figure}

\noindent
The matrix element of the hadronic current

\begin{eqnarray}
J^{em}_{\nu} \ \equiv \ J^{em}_{\nu}\left(q_1,...,q_n\right) \ \equiv \ 
 <h(q_1),...,h(q_n)|J^{em}_{\nu}\left(0\right)|0>
\end{eqnarray}


\noindent
has to be parameterized by form factors to be discussed below. For the
two pion case the current 

\begin{eqnarray}
J_{\nu}^{em,2\pi} 
 = \left(q_{\nu}^{+}-q_{\nu}^{-}\right)F_{2\pi}\left(Q^2\right)
\end{eqnarray}

\noindent
is determined by only one function, the pion
form factor \(F_{2\pi}\).

For the three pion case the matrix element of the electromagnetic
current is restricted by current conservation and negative parity to the
form

\begin{eqnarray}
 J_{\nu}^{em,3\pi} =
 \epsilon_{\nu\alpha\beta\gamma}\ q_{+}^{\alpha}q_{-}^{\beta}q_{0}^{\gamma}
  \ F_{3\pi}\left(q_{+},q_{-},q_{0}\right)
\end{eqnarray}

\noindent
and $F_{3\pi}$ is dominated by the $\omega$ resonance. The matrix element
for the four pion case will be discussed in sections 3 and 4.



The differential rate can be cast into a product of a leptonic and a
hadronic tensor and a corresponding separation of the phase space

\begin{eqnarray}
d\sigma = \frac{1}{8s} \ L_{\mu\nu} H^{\mu\nu}\  
    d\Phi_2(p_{+}+p_{-};Q,k)\  d\bar\Phi_n(Q;q_1,\dots,q_n) \ 
 \frac{dQ^2} {2\pi} \ , 
\labbel{SIG}
\end{eqnarray}

\noindent
where \(d\bar\Phi_n(Q;q_1,\dots,q_n)\) denotes the \(n\) body phase space
 with all statistical factors coming from the hadronic final state included.

The leptonic tensor $L_{\mu\nu}$ is process independent, the modeling
of hadronic physics enters the tensor $H_{\mu\nu}= J^{em}_\mu (J^{em}_\nu)^*$
only. The leptonic tensor is symmetric, and hence it is only the real
symmetric part of $H_{\mu\nu}$ which enters. It has the following form
 (\(J^{em}_\mu\) current conservation was used to get the expression)

\begin{eqnarray}
L^{\mu\nu} = 
  \frac{-4}{(kp_{-})(kp_{+})} 
  \left[ \ g^{\mu\nu} \  \left(
  p_{-}p_{+}  
 + \frac{(kp_{+})^2+ (kp_{-})^2}{Q^2} \right)
 + \ \ p_{+}^{\mu}p_{+}^{\nu}+p_{-}^{\mu}p_{-}^{\nu} \ \right] 
 \frac{1}{Q^2}
 \ .
\end{eqnarray}


The collinear and soft photon singularities proportional to $1/(1\pm
\cos\theta_\gamma)$ and $1/E_\gamma$ are evident from these expressions,
as well as the $1/Q^2$ enhancement for small $Q^2$. After integrating
the hadronic tensor $H_{\mu\nu}$ over the hadronic phase space one gets

\begin{eqnarray}
 \int \ J^{em}_\mu (J^{em}_\nu)^*  \ \ d\bar\Phi_n(Q;q_1,\dots,q_n) = 
 \frac{1}{6\pi} \left(Q_{\mu}Q_{\nu}-g_{\mu\nu}Q^2\right) \ R(Q^2)
\labbel{rr}
\end{eqnarray}
 
\noindent
where \(R(Q^2)\) is \(\sigma(e^+e^-\rightarrow hadrons)/\sigma_{point}\).

\begin{table}
\begin{center}
$$
\begin{array}{||c||c||c||c|c|c||}
\hline
&
&
&
\multicolumn{3}{|c|}{{\rm Event~rates}}\\
\hline
%\cline{4-6}
{\rm Collider }& \sqrt{s}
&{\rm Integrated~luminosity}, {\rm fb}^{-1} & \theta_{\rm min} = 5^\circ 
& \theta_{\rm min} = 7^\circ   &  \theta_{\rm min} = 10^\circ 
\\ \hline \hline
{\rm DAPHNE} & 1.02 & 1 & 13\cdot 10^6 & 12 \cdot 10^6 
&  10 \cdot 10^6 \\ \hline \hline
B-{\rm factory} & 10.6 & 100 & 4 \cdot 10^6 & 3.5 \cdot 10^6 
&  3 \cdot 10^6 \\ \hline \hline
{B-\rm factory} & 10.6 & 100 & 2.7\cdot 10^6 & 2.3\cdot 10^6
&  2.0\cdot 10^6 \\ 
\hline
\end{array}
$$
\vspace*{0.5cm}
\caption{Estimated number of  radiative 
events $e^+e^- \to \ hadrons \ + \ \gamma$ for different center of mass
energies from \Eq{1}. In the first two rows by \(hadrons\) we mean
\(\pi^+\pi^-\) and 
the minimal photon energy is $0.1$ GeV. The third row is 
 obtained for a continuum
 contribution in the region
  \(2 \  {\rm GeV} < \sqrt{Q^2} < 3.7 \ {\rm GeV}\)
 assuming a constant \(R=2.4\).  }
\end{center}
\label{t1}
\end{table}


 After additional integration over the photon angles
 (azimuthal angle is integrated over the full range, while
the polar angle within \(\theta_{min} < \theta < \pi-\theta_{min}\) )
 in  \Eq{SIG}
  the differential distribution 

\begin{eqnarray}
Q^2 \frac {{\rm d}\sigma}{{\rm d} Q^2} = \frac
{4 \alpha^3}{3 s} R(Q^2)
\left \{ \frac {(s^2+Q^4)}{s(s-Q^2)} 
\log \frac {1+\cos \theta_{\rm min}}{1-\cos \theta_{\rm min}}
-\frac {(s-Q^2)}{s}\cos \theta_{\rm min} \right \}
\labbel{1}
\end{eqnarray}


\noindent
can be used to calculate the event rate observed for realistic photon
energy and angular cuts (see Tab.1).

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
\section{Isospin relations.}

The emphasis of this paper is towards hadronic final states consisting of
 four pions and a photon. Before entering a discussion of a model 
 dependent parameterization of the form factors (see section 5) we recall
 the constraints coming from isospin invariance. They relate the amplitudes
 of the \(e^+ e^- \to 2\pi^+2\pi^-\) and \(e^+ e^- \to \pi^+\pi^- 2\pi^0\)
 processes and those for \(\tau\) decays into \(\pi^-3\pi^0\) and
 \(\pi^+2\pi^-\pi^0\). The amplitude of the \(\tau\) decay into arbitrary
 number of hadrons plus a neutrino is given by

\begin{eqnarray}
 {\cal M}_{\tau} = \ \ \frac{G_F}{\sqrt{2}} \ \cos\theta_c \ \
 \bar v\left(p_{\nu}\right) \gamma^{\alpha}\left(1-\gamma_5\right)
 u\left(p_{\tau}\right) \ \ J_{\alpha}^{-} \ ,
\end{eqnarray}

with 

\begin{eqnarray}
 &&J^{-}_{\alpha} \ \equiv \ J^{-}_{\alpha}\left(q_1,...,q_n\right) \ \equiv \ 
 <h(q_1),...,h(q_n)|J^{-}_{\alpha}\left(0\right)|0> \nonumber \\
 {\kern-30pt} {\rm and} \ \ \ \ \
 &&J^{-}_{\alpha}(0) = \bar d \ \gamma_\alpha \ u
 \ \ \ .
\end{eqnarray}
We use the same letter $J$ for the operator and its matrix element and
restrict our considerations to the Cabbibo allowed vector part of the
hadronic current.



\noindent
It leads to the differential distribution

\begin{eqnarray}
\frac{d\Gamma}{dQ^2} = 2 \ \Gamma_e \frac{\cos^2\theta_c}{m_{\tau}^2}
 \left(1-\frac{Q^2}{m_{\tau}^2}\right)^2 
 \left(1+2\frac{Q^2}{m_{\tau}^2}\right) R^{\tau}\left(Q^2\right) \ ,
\labbel{diff}
\end{eqnarray}

with 

\begin{eqnarray}
 \int \ J_\mu^{-} J^{-*}_\nu  \ \ d\bar\Phi_n(Q;q_1,\dots,q_n) = 
 \frac{1}{3\pi} \left(Q_{\mu}Q_{\nu}-g_{\mu\nu}Q^2\right) \ R^{\tau}(Q^2) \ .
\labbel{rrt}
\end{eqnarray}
Note the relative factor of 2 between the definitions in \Eq{rr} and 
 \Eq{rrt}.

Ignoring the issues of isospin breaking and radiative corrections,
the electromagnetic current can be decomposed into an isospin singlet piece
and a part transforming like the third component of an isospin triplet:

\ba
 J^{em} = \frac{1}{\sqrt{2}}\ \  J^3 + \frac{1}{3\sqrt{2}}\ \  J^{{\rm I}=0}
\ea

\noindent
 whereas the charged current generating \(\tau\) decays is given by

\ba
J^- = \frac{1}{\sqrt{2}} \ \ (J^1-i \ J^2) \ \ .
\ea

Final states with an even number of pions are produced through the isospin
 one part only, whence

\begin{eqnarray}
 \sqrt{2} \ \ J^{em}_{\mu}(2\pi) = J^{-}_{\mu}(2\pi)
 \labbel{rel0}
\end{eqnarray}

\noindent
and \(R(Q^2)=R^{\tau}(Q^2)\) for two pion final states.
 
 A similar relation for the four pion final state is easily obtained 
 as follows: the transition from the vacuum to four pions is mediated
 through a current \(\vec J_{\mu}\left(0\right)\) of the form

\ba
\vec J_{\mu}\left(0\right) =
 \frac{1}{4} \ \int d^4q_1 d^4q_2 d^4q_3 d^4q_4 \ \
   J_{\mu} \ 
   \left(\vec \pi_1\cdot\vec \pi_2 \right)\ \
 \left(\vec \pi_3
 \times \vec \pi_4 \right) \ ,
 \labbel{curr}
\ea

\noindent
where \(\vec\pi_i \equiv \vec\pi(q_i)\) denotes the pion field in the
 Cartesian basis. 
The letter $J$ is again used both for the operator and the function in the
integrand which corresponds essentially to the transition amplitude.
The function 
 \(J_{\mu} \equiv J_\mu(q_1,q_2,q_3,q_4) \) is symmetric (anti-symmetric)
 with respect to the interchange of \(q_1\) and \(q_2\) (\(q_3\) and \(q_4\)).
 The combination of pion fields relevant for the transition to four pions 
 with total charge -1 and 0 respectively is given by
\ba
 J^{-}_{\mu}\left(0\right) &=&
 \frac{1}{4} \ \int d^4q_1 d^4q_2 d^4q_3 d^4q_4 \ \
    J_{\mu}\cdot
 \left( \pi^+_1 \pi^-_2+ \pi^-_1 \pi^+_2
       + \pi^0_1 \pi^0_2\right) \left( \pi^-_3 \pi^0_4- \pi^-_4 \pi^0_3\right)
\nonumber \\
 J^{3}_{\mu}\left(0\right) &=&
 \frac{1}{4} \ \int d^4q_1 d^4q_2 d^4q_3 d^4q_4 \ \
    J_{\mu}\cdot
  \left( \pi^+_1 \pi^-_2+ \pi^-_1 \pi^+_2
       + \pi^0_1 \pi^0_2\right)\left( \pi^+_3 \pi^-_4- \pi^-_3 \pi^+_4\right)
 \ . 
\ea

Taking the matrix element of these operators between vacuum and the states


 \noindent
\(\langle \pi^+(p^+)\pi^-(p^-)\pi^0(p_1)\pi^0(p_2)|\) etc. one immediately
 arrives at

\ba
\langle \pi^+ \pi^- \pi_1^0 \pi_2^0 | J^3_{\mu} | 0 \rangle &=& 
J_{\mu}(p_1,p_2,p^+,p^-) \nonumber \\
\langle \pi^+_1 \pi^+_2 \pi^-_1 \pi^-_2 | J^3_{\mu} | 0 \rangle &=&
J_{\mu}(p_2^+,p_2^-,p_1^+,p_1^-) +
J_{\mu}(p_1^+,p_2^-,p_2^+,p_1^-) \nonumber \\
&&{\kern-15pt}+J_{\mu}(p_2^+,p_1^-,p_1^+,p_2^-)+
 J_{\mu}(p_1^+,p_1^-,p_2^+,p_2^-)\nonumber \\
\langle \pi^- \pi^0_1 \pi^0_2 \pi^0_3 | J^{-}_{\mu} | 0 \rangle &=& 
J_{\mu}(p_2,p_3,p^-,p_1)+
J_{\mu}(p_1,p_3,p^-,p_2)+J_{\mu}(p_1,p_2,p^-,p_3)
\nonumber \\
\langle \pi^-_1 \pi^-_2 \pi^+ \pi^0 | J^{-}_{\mu} | 0 \rangle &=& 
J_{\mu}(p^+,p_2,p_1,p^0)+
J_{\mu}(p^+,p_1,p_2,p^0)
 \labbel{rel} \ ,
\ea

\noindent
which connects \(\tau\) decay and electron positron annihilation.
It is clear from \Eq{rel}
 that only one matrix element, namely the one for \((+\ -\ 0\ 0)\) ,
 needs to be programmed and the remaining ones can be obtained by relabeling
 arguments. Interference terms between the two partitions \cite{Pais,PS,Gil}
  (3,1) and (2,1,1) are present in the differential distributions.
 For the integrated rates induced by the currents one obtains

\ba
 R\left(+\ - \ 0 \ 0\right) &=& \frac{1}{2} \ A 
 {\kern+18pt} \ \ \ ; \ \ \ 
 R^{\tau}\left(-\ - \ + \ 0\right) =  A + \frac{1}{2} \ B \nonumber \\
 R\left(+\ + \ - \ -\right) &=&  A \ +  \ B \ \ \ ; \ \ \ 
   R^{\tau}\left(-\ 0 \ 0 \ 0\right) {\kern+9pt}
 = \frac{1}{2}\left( A \ +  \ B\right) \ ,
\ea 

\noindent
with

\ba
 A &=& -\frac{2\pi}{Q^2}\int \ J^\mu\left(q_1,q_2,q_3,q_4\right)
             J^{*}_\mu \left(q_1,q_2,q_3,q_4\right)
    \ \ d\bar\Phi_4(Q;q_1,\dots,q_4)\nonumber \\
 B &=& -\frac{4\pi}{Q^2}\int \ {\rm Re}\left(J^\mu\left(q_1,q_2,q_3,q_4\right)
             J^{*}_\mu \left(q_1,q_3,q_2,q_4\right) \right)
    \ \ d\bar\Phi_4(Q;q_1,\dots,q_4) \ ,
\ea

\noindent
consistent with the familiar relations between \(\tau\) decays and
 \(e^+e^-\) annihilation into four pions:

\ba
 R^{\tau}\left(-\ 0 \ 0 \ 0\right) &=&  \frac{1}{2}\  R\left(+\ + \ - \ -\right)
\nonumber \\
 R^{\tau}\left(-\ - \ + \ 0\right) 
 &=& \frac{1}{2} \ R\left(+\ + \ - \ -\right) \ 
 + \ R\left(+\ - \ 0 \ 0\right) \ .
\labbel{CVC}
\ea
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
\section{The \(\pi^{\pm}\)-\(\pi^0\) mass difference 
 and the isospin relations.}

 All the relations obtained in the previous section are strictly
 applicable only
 in case all pions in the final states have the same mass, which is
 obviously not true. The relatively large (about 3.6\%) 
\(\pi^{\pm}\)-\(\pi^0\) mass difference will affect the 
 \(R(Q^2)\leftrightarrow R^\tau(Q^2)\) relation even if the relations
 \Eq{rel0} and \Eq{rel} still hold. The CVC hypothesis 
 and the assumption that 
transitions to
an even number of pions in the final state are described
by the iso-vector current
are well established experimentally \cite{EI}.
 It is thus natural to assume that the relations between currents hold
 and the question is up to what precision we can ignore the 
 \(\pi^{\pm}\)-\(\pi^0\) mass difference. This should give at least 
 an indication of the size of these "kinematic" isospin violations.
 We will address this issue
 here for two and four pion final states. 

 First of all for two pion final state in the case of unequal masses the
 relations between the currents should be changed from 

\ba
 J^{em}_\mu(2\pi) = \sqrt{2} \ J^-_\mu(2\pi) = (q_{1,\mu}-q_{2,\mu}) \ F(Q^2) \ ,
\ea 

\noindent
(which is obviously not conserved) to

\ba
 J^{em}_\mu(2\pi) = \sqrt{2} \ J^-_\mu(2\pi) = \left(q_{1,\mu}-q_{2,\mu} 
   - \frac{q_{1,\mu}+q_{2,\mu}}{Q^2}\ Q.(q_1-q_2)\right) \ F(Q^2) \ ,
\ea 

\noindent
where \(F(Q^2)\) is the pion form factor. For the relation between
  \(R(Q^2)\)  and \(R^\tau(Q^2)\) 
 one finds

\ba
 R^\tau(Q^2)= \frac{\lambda^{\frac{3}{2}}\left(Q^2,m_-^2,m_0^2\right)}
 {\lambda^{\frac{3}{2}}\left(Q^2,m_-^2,m_-^2\right)} R(Q^2) \ ,
\ea 

\noindent
where \(m_-\) (\(m_0\)) is the \(\pi^\pm\) (\(\pi_0\)) mass and
\(\lambda(a,b,c)=a^2+b^2+c^2-2ab-2ac-2bc\). As a result
we have

\begin{eqnarray}
&&{\kern-40pt}\frac{1}{\Gamma_e} \ \
\frac{d\Gamma(\tau^-\to \nu_\tau\pi^-\pi^0)}{dQ^2} = \nonumber \\
   &&\frac{3\cos^2\theta_c} {2\pi \alpha^2 m_{\tau}^2}
 \ \ Q^2 \ \left(1-\frac{Q^2}{m_{\tau}^2}\right)^2 
 \left(1+2\frac{Q^2}{m_{\tau}^2}\right) 
 \frac{\lambda^{\frac{3}{2}}\left(Q^2,m_-^2,m_0^2\right)}
 {\lambda^{\frac{3}{2}}\left(Q^2,m_-^2,m_-^2\right)}
 \ \sigma(e^+e^-\to\pi^+\pi^-) \ .
 \labbel{CVCc}
\end{eqnarray}

 
 The integral over \(Q^2\) has been calculated
 with and without
 the additional factor 
\( {\cal F}= \lambda^{\frac{3}{2}}\left(Q^2,m_-^2,m_0^2\right)\) /
 \({\lambda^{\frac{3}{2}}\left(Q^2,m_-^2,m_-^2\right)}\) using the 
  pion form-factor of \cite{BKM}. For the ratio of the two results
 one finds

\ba
 {\cal R} \equiv \frac{\Gamma(with \ {\cal F})}{\Gamma(without \ {\cal F})}
 = 1.0094 \ .
\ea

\noindent
 This leads to a slight enhancement of the \(\tau\) decay rate
 compared to the CVC prediction with equal pion masses.
Taking now the prediction
\cite{EI} for the branching ratio based on
 \(e^+e^-\) data
  which amounts to \((24.52 \ \pm \ 0.33)\%\)
and scaling it by \({\cal R}\) 
 one finds for the modified
  CVC prediction 
 \( (24.75 \ \pm \ 0.33)\% \) to be compared with the Particle Data
 Group value \cite{PDG} of \((25.32 \ \pm \ 0.15)\%\).
 The difference between the measurement and the CVC prediction
 is reduced from 2.2 \(\sigma\) 
 to 1.6 \(\sigma\).

\begin{figure}[htbp]
\epsfig{figure=dif_m0.ps, width=0.75\textwidth,height=6.5cm}
\caption{The ratio
 \( {\cal R}_d \equiv  \ \frac{\sigma_1-\sigma_2}{\sigma_2}\), where
 \(\sigma_1\) (\(\sigma_2\)) is \(e^+e^-\to 2\pi^0 \pi^-\pi^+\) cross section
 calculated for true pion masses (with all masses equal to \(m_-\)).
}
\label{f8}
\end{figure}

 The situation is more complicated for the four pion case. All \(4\pi\)
 modes have different numbers of \(\pi^0\), whence the phase space 
 is different for each of the mode. Moreover, for a comparison between
 \(R(Q^2)\) and \(R^\tau(Q^2)\) one has to integrate over four particle
 phase space and it is impossible to obtain a simple analytical result.
  To estimate a size of the effect we 
  integrated the quantities \(R(--+0)\) etc. according to \Eq{diff}
 using 
 the current discussed in the next section once
 assuming that all masses are equal to \(m_-\) 
 and once taking the real masses. 
 For the mass corrections of the integrals we find
 
 \ba
 (-000)\ : \ 5.0  \% \ \ \ ; \ \ \ (--+0)\ : 2.4 \% \ \ \ ;\ \ \ 
 (++--)\ : \ 0 \% \ \ \ ; \ \ \ (+-00)\ : 4.6 \%
 \ea

 Mass effects alone thus modify the integrated version of the \Eq{diff} to

 \ba
 \frac{1}{1.050}\ \Gamma(-000) &=& \frac{1}{2} \ \Gamma(++--) \nonumber \\
 \frac{1}{1.024}\ \Gamma(--+0) &=& \frac{1}{2} \ 
  \Gamma(++--) + \frac{1}{1.046}\ \Gamma(+-00)
 \ ,
 \ea

\noindent
where 

\ba
 &&\Gamma(-000)=\Gamma(\tau^-\to \nu_\tau \pi^- 3\pi^0) \ \ ; \ \
      \Gamma(--+0)=\Gamma(\tau^-\to \nu_\tau 2\pi^- \pi^+ \pi^0) \nonumber \\
 &&\Gamma(++--)=  2 \ \Gamma_e \frac{\cos^2\theta_c}{m_{\tau}^2} \int
 \left(1-\frac{Q^2}{m_{\tau}^2}\right)^2 
 \left(1+2\frac{Q^2}{m_{\tau}^2}\right) R(++--) \ dQ^2 \nonumber \\
 &&\Gamma(+-00)=  2 \ \Gamma_e \frac{\cos^2\theta_c}{m_{\tau}^2} \int
 \left(1-\frac{Q^2}{m_{\tau}^2}\right)^2 
 \left(1+2\frac{Q^2}{m_{\tau}^2}\right) R(+-00) \ dQ^2 \ .
\ea

In principle this correction depends on the form of the current which
  will be specified in the next section.
  However, since \(a_1\) and \(\omega\) dominance
 and the qualitative behaviour of the matrix element
 is well established
  this result will be applicable also to other forms of the current.



 It is also instructive to consider the mass effects on the differential
 rate for the \(e^+e^- \to 2\pi^0 \pi^+\pi^-\) process.
In Fig.\ref{f8} we plot the ratio 
 \({\cal R}_d \equiv \frac{\sigma_1-\sigma_2}{\sigma_2}\), where
 \(\sigma_1\) (\(\sigma_2\)) is the \(e^+e^-\to 2\pi^0 \pi^-\pi^+\)
  cross section
 calculated for true pion masses (with all masses equal to \(m_-\)).
 The difference amounts again
 up to a few percent.
The relations between the differential
  \(\tau\) decay rates and the \(e^+e^-\) cross sections obtained
  in the previous section will be violated at that level of accuracy even
  if  \Eq{rel} holds. However, to test these predictions experimentally
more precise measurements of the \(e^+e^-\to 4\pi\) cross section are required,
  where now a typical systematic error amounts to roughly 15 \%.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
\section{The hadronic current. }


 As one can see from section 3 it is enough to construct only
 the hadronic current for the \((+-00)\) mode, while the
 other ones can be obtained using the relations \Eq{rel}.
 Its construction was
 based on  \cite{Fink} with some small changes allowing for 
 preserving the relations \Eq{rel}. The basic building block
 of the current contains a part built on the assumption of  \( a_1\) vector
 dominance plus an \(\omega\) exchange contribution.
 However only by adding an \(f_0\)
  contribution one
 can recover the proper chiral limit \cite{FWW}.
  The complete current can be written as
 a sum of these three contributions
 
\begin{eqnarray}
 \Gamma^{\mu}_{\rho^0 \rightarrow 2\pi^0 \pi^+ \pi^-} =
 \Gamma^{\mu}_{a_1} +\Gamma^{\mu}_{f_0}+\Gamma^{\mu}_{\omega} \ .
 \labbel{sum}
\end{eqnarray}

\begin{figure}[ht]
\begin{center}
\epsfig{file=Fig2.ps,width=14cm,height=5cm}
\end{center}
\caption{Diagrams contributing to the hadronic current.
}
\label{f2}
\end{figure}


They are depicted schematically in Fig.\ref{f2} 
 and described in detail in the Appendix. Here we present
 some numerical tests of the current and a
 comparison between results obtained using the current \Eq{sum}
 and experimental data. One should add that the parameters of
 the model are kept as in \cite{Fink} even if in principle they
 should be re-fitted as the current is a bit different from the
 original one of \cite{Fink} and new and improved data have became
 available. This, however, is beyond the scope of this paper.  

 Starting from tests of the code of the current,
 first one can check if the Monte Carlo program
 reproduces the known \cite{tauola} analytical results
 of the partial \(\tau\) decay
 widths in the chiral limit

\ba
 \gamma_1 \equiv 
 \frac{\Gamma\left(\tau\to\nu_{\tau}2\pi^- \pi^+ \pi^0\right)}{\Gamma_e}
 = \frac{\cos^2\theta_c}{15} \left(\frac{m_\tau}{2\pi f_{\pi}}\right)^4
 \frac{1}{128}\left(\frac{1009}{96}-\pi^2\right) 
\labbel{width}
\ ,
\ea

\ba
 \gamma_3 \equiv 
 \frac{\Gamma\left(\tau\to\nu_{\tau}\pi^- 3\pi^0\right)}{\Gamma_e}
 = \frac{\cos^2\theta_c}{15} \left(\frac{m_\tau}{2\pi f_{\pi}}\right)^4
 \frac{1}{256}\left(\pi^2-\frac{437}{48}\right) \ .
\ea

\begin{table}
\begin{center}
$$
\begin{array}{|c|c|c|c|}
\hline
 {\rm Decay \  mode} 
& {\rm MC \ result \  (direct)}
& {\rm MC \ result \ (CVC)}
& {\rm analytical \  result}\\
\hline
%
 \gamma_1
&0.026786(4)
&0.026786(4)
&0.026788 \\
\hline
 \gamma_3
&0.015998(3)
&0.015996(2)
&0.015999 \\
\hline
\end{array}
$$
\vspace*{0.5cm}
\caption{
Comparison between analytical and Monte Carlo results in the chiral limit. 
 }
\end{center}
\label{t2}
\end{table}

 \Eq{width} differs from Eq.(38) of \cite{tauola}.
It was not discovered there that the analytical result is wrong as the tests
 were done at 0.2 \% (one sigma) precision level and the difference amounts
 to 0.34 \% . Decay widths can be obtained in two different ways. One way
 is just their direct calculation. The second one is by using the known
 relations to the \(e^+e^-\to 2\pi^+ 2\pi^- \) and 
 \(e^+e^-\to \pi^+\pi^- 2\pi^0\) cross sections \cite{Tsai,Fink}.
 The results of the numerical tests are summarized in 
 Tab.2, where CVC means the decay width was obtained through its relation
 to the \(e^+e^-\) cross sections.
As one can see the results of the test performed at 0.02 \% accuracy 
 level are quite satisfactory. 
 This agreement gives confidence in the numerical stability of our
 program.


\begin{table}
\begin{center}
$$
\begin{array}{|c|c|c|c|}
\hline
 {\rm Mode} 
& {\rm \cite{Fink}}
& {\rm present \  model}
& {\rm experiment}\\
\hline
%
 {\rm Br}\left( \tau^-\to\nu_{\tau}2\pi^-\pi^+\pi^0 \right)
&3.11\%
&4.33\%
&4.22(10)\% \\
\hline
 {\rm Br}\left( \tau^-\to\nu_{\tau}\pi^-
  \omega\left(\pi^-\pi^+\pi^0\right) \right) 
&1.20\%
&1.48\%
&1.73(14)\% \\
\hline
%
 {\rm Br}\left( \tau^-\to\nu_{\tau}\pi^-3\pi^0 \right)
&0.98\%
&1.14\%
&1.11(14)\% \\
\hline
\end{array}
$$
\vspace*{0.5cm}
\caption{
 Branching ratios of \(\tau\) decay modes. Results of \cite{Fink} 
 and the present current are compared to experimental data. 
 }
\end{center}
\label{t3}
\end{table}

Now we can test the physical predictions of the current. Let us start
 with \(\tau\) decay branching ratios. The results are
 summarized in Tab.3, where we put for completeness  also the results 
 presented in \cite{Fink} and the experimental results are taken from
 \cite{PDG}. The agreement of the predictions of the current \Eq{sum}
 with the experimental data is satisfactory for 
 \(\tau^-\to\nu_{\tau}\pi^-3\pi^0\) decay mode.
Comparing however the results for the  
 \(\tau^-\to\nu_{\tau}2\pi^-\pi^+\pi^0\) and
  \(\tau^-\to\nu_{\tau}\pi^-
  \omega\left(\pi^-\pi^+\pi^0\right)\)
  modes
 it seems that the \(\omega\)
 part of the current does not represent the data well, even if
 the total branching ratio for the \(\tau^-\to\nu_{\tau}2\pi^-\pi^+\pi^0\)
 decay mode
 agrees with the data. Again the results of the partial
 decay widths were obtained as in the chiral limit by direct calculation
 and checked by relating them to the simulated \(e^+e^-\) cross sections.
 Agreement was found within statistical errors  proving that the
 code of the current fulfills the CVC relations \Eq{CVC}
 (if integrals are performed with \(m_-=m_0\)). 



\begin{figure}[htbp]
\epsfig{figure=xs4pi.eps, width=0.45\textwidth,height=7cm}
\hfill
\epsfig{figure=xs4pi_we.eps,width=0.414\textwidth ,height=6.914cm}
\caption{
Comparison of data (left figure, see text) and predictions (right figure) 
obtained using the current \Eq{sum}
 (filled squares) and those of \cite{Fink} (empty squares) respectively
for $\sigma(e^+e^- \to 2 \pi^+2\pi^-)$.
}
\label{f3}
\end{figure}

\begin{figure}[htbp]
\epsfig{figure=xs2pi.eps, width=0.45\textwidth,height=7cm}
\hfill
\epsfig{figure=xs2pi_we_go.eps,width=0.414\textwidth ,height=6.914cm}
\caption{
Comparison of data (left figure, see text) and predictions (right figure) 
obtained using the current \Eq{sum}
 (filled squares) and those of \cite{Fink} (empty squares) respectively
for $\sigma(e^+e^- \to \pi^+\pi^-2\pi^0 )$.
}
\label{f4}
\end{figure}

In the next step the predictions for the \(e^+e^-\to 2\pi^+ 2\pi^-\)
 (Fig.\ref{f3}) and \(e^+e^-\to \pi^+\pi^-2\pi^0\) (Fig.\ref{f4})
  cross sections
 are compared with data.
The plots taken from \cite{CMD2} 
contain data sets from OLYA \cite{OLYA}, ND \cite{ND1,ND2},
 CMD \cite{CMD}, SND \cite{SND}, CMD2 \cite{CMD2} plus results
 of the Orsay \cite{O1,O2,O3} and Frascati \cite{Fr1,Fr2,Fr3} groups.
 The data have a typical systematic error
 of about 15\% (shown in plots only for some of the data sets) and
 we have thus decided not to refit the parameters entering
 the current as the agreement between the Monte Carlo and the data
 is acceptable. The modification of the current of \cite{Fink}
 we performed leads to significantly better agreement between the theoretical
 predictions and the data in the mode \(e^+e^-\to \pi^+\pi^-2\pi^0\).
 Considering the agreement between the \(\tau\) decay rates and
 the prediction we
 conclude that the model
 reproduces the data well even if the description of the \(\omega\) part
 of the current is not completely satisfactory.




%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{The Monte Carlo program.}

 The idea behind the structure of the Monte Carlo program
 is to allow for a simple addition of new final state modes into the program
 and for a simple replacement of the current(s) of the existing modes.
 The program thus exhibits a modular structure. For
 the generation of the four momenta of the mesons no sophisticated method
 of a variance reduction was applied which is efficient
 only in case of a well defined distribution.
 The process to be simulated by the program in its final stage is
 \(e^+ e^- \rightarrow \gamma \ + \ hadrons \) 
 with an exclusive description of final states,
even if till now
 only  \(\pi^+ \pi^-\), \(2\pi^0 \pi^+ \pi^-\) and \(2\pi^+ 2\pi^-\)
 hadronic final states are implemented. The LL radiative QED
 corrections were taken into account using structure function method
 as developed in \cite{CCR} and limited to the initial emission only.
 In fact the program can run in one of two modes (chosen by a user)
  one with collinear radiation
 and one without it.
 Hard large angle photon 
 emission is limited to initial state radiation,
  which is justified by \cite{BKM} where it was demonstrated for the 
 \(\pi^+ \pi^-\) hadronic state that the contribution from the final
 state emission as well as the initial-final state interference can be
 reduced to a negligible level by applying suitable cuts.
 
 The generation of the multi particle phase space is based on the
 following representation of a Lorentz invariant phase space

 \begin{eqnarray}
 &&{\kern-15pt}{\rm dLips}_{n+1}\left(Q,k,q_1,...,q_n\right)=
 \frac{1}{\left(2\pi\right)^{n-1}} \ \ {\rm dQ}_1^2 \ ... \ {\rm dQ}_{n-1}^2 
   \ {\rm dLips}_2\left(Q,Q_1,k\right) 
 \ {\rm dLips}_2\left(Q_1,Q_2,q_1\right)
 \nonumber \\
 &&{\kern+120pt}\ \ ... \ \ {\rm dLips}_2\left(Q_{n-2},Q_{n-1},q_{n-2}\right)
 {\rm dLips}_2\left(Q_{n-1},q_{n-1},q_n\right) \ ,
 \end{eqnarray}  

\noindent
where \(Q\) is a total four momentum of the \(photon \ + \ hadrons\) state
(it is not equal to the sum of the initial \(e^+ e^-\) four momenta as we
 allow for an additional initial collinear emission), \(k\) is the photon
 four momentum, \( q_1\ ... \ q_n\) are the four momenta of the hadrons,

 \begin{eqnarray}
 Q_1 = q_1 + \  ... \ +q_n \ \ {\rm and} \ \ Q_i = Q_{i-1}-q_{i-1} 
 \ \ {\rm for} \ \
 i=2,\ ... \ n-1 \ \ ,
 \end{eqnarray}  
 
 and \({\rm dLips}_2\left(k_1,k_2,k_3\right)\) is a two body phase space

 \begin{eqnarray}
 {\rm dLips}_2\left(k_1,k_2,k_3\right) = \frac{1}{32\pi^2} \ 
 \frac{\lambda^{\frac{1}{2}}\left(k_1^2,k_2^2,k_3^2\right)}{k_1^2} \ 
 {\rm d}\Omega_3
 \end{eqnarray}

 with   \(\lambda\left(a,b,c\right)= a^2+b^2+c^2-2ab-2ac-2bc\) and 
 \({\rm d}\Omega_3\) the \({\bf k}_3\) solid angle.
  
 The generation flow is the following:
first the collinear radiation is generated and a four momentum \(Q\)
 of the visible final state is calculated and then boosted to its
 rest frame(RF) (a CM of the visible final state). This part is omitted
 in the mode running without collinear emission and then \(Q\) is a sum
 of the electron and positron four momenta.
 In the second step the visible hard photon four momentum is generated.
 Its energy is generated flat even if the energy distribution is govern
 by a competition between soft \(\sim \frac{1}{E_\gamma}\) and a
 complicated resonant spectrum which depends on details of the hadronic
 current. Of course this way the generator is not extremely efficient,
 but it is more universal and will work equally well with all reasonable
 modifications of the hadronic current. The photon polar angle is
 generated in \(Q\) RF with the distribution accounting 
 for collinear emission peaks,
 which are the same for all hadronic modes 
 as the initial state is ever the same.
 Its azimuthal angle again is generated with a flat distribution in \(Q\) RF.
   Then a chain of a generations
 of \(Q_i^2\ , \ i=2,...,n-1\) follows (\(Q_1\) is fixed when we generate
 the photon four momentum). 
 They are generated flat within their allowed limits

 \begin{eqnarray}
  \left(\sum_{k=i}^{n}m_k\right)^2\ < \ Q_i^2 \ 
  < \ \left(Q_{i-1}^{(0)}-m_{i-1}\right)^2
  \end{eqnarray}

where \(Q_i^{(0)}\) is a zeroth component of the \(Q_i\) four momentum.
At the end a generation of the solid angles \(\Omega_i,\ i=1,...,n-1\)
 follows. They are generated flat ( \( \Omega_i \) in \(Q_i\) RF).
 All generated four momenta are then transformed with the use of the proper
 boosts and rotations into the CM system of the initial \(e^+e^-\) particles. 
 The distribution of the events \(\sim |{\cal M}|^2\), where \({\cal M}\) is
 a matrix element of a given process is obtained by means of the 
 'hit or miss' method. The cross section is calculated using \Eq{SIG}
 both for weighted and unweighted event samples. The unweighted events
 are stored in a file when requested.

 Let us now discuss the cuts, which
 reduce the contribution of the final state emission to the cross section 
 to a negligible level allowing thus for extraction of \(R(Q^2)\) from
  the measurement of the
 \(e^+e^-\to \ hadrons \ + \gamma\) cross section.
 We recall that in \cite{BKM} it was shown that the following set of 
 angular cuts 
 
\ba
 {\rm cuts1:} \ \ \ \ \ \ \ \ \ \
 \left( 7^{\circ} < \theta_\gamma < 20^{\circ} 
 \ \ \ \ \ or \ \ \ \ \ 160^{\circ} < \theta_\gamma < 173^{\circ}\right) 
 \ \ \  and 
  \ \ \ 30^{\circ} < \theta_\pi < 150^{\circ} \ ,
 \labbel{cuts}
\ea

\noindent
 ( \(\theta_\gamma\) (\(\theta_\pi\) ) is the photon (pion) polar angle)
 fulfils this requirement for the 
 \(e^+ e^- \to \pi^+ \pi^- \gamma\) cross section. It reduces, however,
 the observed cross section significantly. This starts to become dramatic, when
 one runs at energies well above 1 GeV. The following
 set of cuts

\ba
 {\rm cuts2:} \ \ \ \ \ \ \ \ \ \
 &&\left( 7^{\circ} < \theta_\gamma < 20^{\circ}  \ \ \
 and  \ \ \ 30^{\circ} < \theta_\pi < 173^{\circ}\right) \nonumber \\
   &&{\kern-40pt}or  \ \ \ 
 \left( 160^{\circ} < \theta_\gamma < 173^{\circ}  \ \ \
 and  \ \ \ 7^{\circ} < \theta_\pi < 150^{\circ}\right) \ ,
\labbel{cuts1}
\ea

\noindent
also reduces the contribution from final state radiation
 to a
 negligible level due to the fact that
 the pions and photon are well separated as in the
 previous case. 


\begin{figure}[htbp]
\epsfig{figure=cuts_eff.ps, width=0.45\textwidth,height=7cm}
\hfill
\epsfig{figure=cuts_0_eff.ps,width=0.45\textwidth ,height=7cm}
\caption{The differential \(e^+e^-\to 4\pi\gamma\) cross sections at
 beam energy 1.5 GeV with minimal photon energy equal 0.1 GeV with
  no cuts on pions angles and \(7^{\circ} < \theta_\gamma < 173^{\circ}\)
 ( no cuts) and two sets of angular cuts \Eq{cuts}
 (with cuts1) and \Eq{cuts1} (with cuts2), where \(q\equiv \sqrt{Q^2}\)
 is an invariant mass of the \(4\pi\) system.
}
\label{f5}
\end{figure}

\begin{figure}[htbp]
\epsfig{figure=cuts_eff_10.ps, width=0.45\textwidth,height=7cm}
\hfill
\epsfig{figure=cuts_0_eff_10.ps,width=0.45\textwidth ,height=7cm}
\caption{The differential \(e^+e^-\to 4\pi\gamma\) cross sections at
 beam energy 5 GeV with minimal photon energy equal 0.2 GeV with
no cuts on pions angles 
 and \(7^{\circ} < \theta_\gamma < 173^{\circ}\)( no cuts) 
 and angular cuts of \Eq{cuts1}
 (with cuts2), where \(q\equiv \sqrt{Q^2}\)
 is an invariant mass of the \(4\pi\) system.
}
\label{f6}
\end{figure}


At the same time the cross section reduction is much
 smaller, especially for higher beam energies and higher energies
 of the observed photons. The effect of the two sets of cuts on the
 cross sections with four pions in the final state is presented 
in Fig.\ref{f5} for  \(2E_{beam}\) = 3 GeV. 
 For higher beam energies the effect of the cross section
 reduction is much higher and at 2 \(E_{beam}\) = 10 GeV the cross
 sections for cuts specified in \Eq{cuts} is reduced almost to zero. 
 For the cuts specified in \Eq{cuts1} the results
 are presented in Fig.\ref{f6} and the reduction remains tolerable.

\begin{table}
\begin{center}
$$
\begin{array}{||c||c||c|c||}
\hline
&
&
\multicolumn{2}{|c|}{{\rm Event~rates}}\\
\hline
%\cline{4-6}
\sqrt{s}
&{\rm Integrated~luminosity}, {\rm fb}^{-1} 
& \ \ \ 2\pi^+2\pi^-\gamma \ \ \ 
& \ \ \ 2\pi^0\pi^+\pi^-\gamma \ \ \ 
\\ \hline \hline
\ 1 \ {\rm GeV}\ & 1 & 1.04\cdot 10^4 & 1.00 \cdot 10^4 
 \\ \hline \hline
\  3\ {\rm GeV}\ & 1 & 4.66 \cdot 10^4 & 2.33 \cdot 10^4 
  \\ \hline \hline
\  10\ {\rm GeV}\  & 100 & 1.86 \cdot 10^5 & 2.17 \cdot 10^5 \\ 
\hline
\end{array}
$$
\caption{Estimated number of  radiative 
events $e^+e^- \to \ 4\pi \ + \ \gamma$ for different center of mass
energies. The minimal photon energy is: 0.05 GeV (first row),
 0.1 GeV (second row), 0.2 GeV (third row). The angular cuts of \Eq{cuts1}
 were applied.
 }
\end{center}
\label{t4}
\end{table}


From Fig.\ref{f6} one concludes that one can measure
 \(R(Q^2)\) at a B factory in the interesting
  region of \( \sqrt{Q^2}\) between 1 GeV and 2.5 GeV  through
  measuring the \(e^+e^-\to 4\pi\gamma\) cross section
 using \Eq{1}. This measurement should have an an accuracy
 much better then 15 \% , which is now the typical
 experimental error in that \(Q^2\) region, 
 allowing for a reduction of the error
 in the calculation of the photon vacuum polarization. 
 From  Table 4 it is clear that the 
  error would be dominated by systematics and not by statistics.





\begin{figure}[htbp]
\epsfig{figure=q2_rad_eff.ps, width=0.45\textwidth,height=6.5cm}
\hfill
\epsfig{figure=q2_0_rad_eff.ps,width=0.45\textwidth ,height=6.5cm}
\caption{The effect of the collinear radiation on the 
 differential \(e^+e^-\to 4\pi\gamma\) cross sections at
 beam energy 0.5 GeV (left picture) and 5 GeV (right picture)
  with minimal photon energy equal to 0.05 and 0.2 GeV 
 correspondingly. In the mode with collinear radiation minimal
 invariant masses of the \(4\pi \ + \ \gamma\) systems  
 of 0.95 GeV and 9.5 GeV were required. 
  \(q\equiv \sqrt{Q^2}\)
 denotes the invariant mass of the \(4\pi\) system.
}
\label{f7}
\end{figure}

 Additional collinear emission ever present in the real experiment
 reduces slightly the cross sections as shown in Figs.\ref{f7} for
 two different modes and two different beam energies.
 Its actual size depends on the cuts on the invariant mass of
 the \(4\pi\ +\ \gamma\) system. The effect is similar for different
 energies and for both charge modes.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Summary.}
A precise value of the cross section for hadron production in electron
positron annihilation at low energies is one of the important
ingredients for a reliable prediction of the anomalous magnetic of the
muon and the electromagnetic coupling at high energies. As an alternative
to a direct measurement at the relevant energy one may use initial state
radiation to reduce the effective energy of electron positron colliders, 
exploiting the large luminosity of ``factories'' and accessing thus a
continuum of hadronic final states.

With this motivation a Monte Carlo generator has been constructed to
simulate the reaction $e^+e^- \to \gamma + 4 \pi$, where the photon is
assumed to be observed in the detector. The hadronic matrix element has
been taken from \cite{Fink}. 
Isospin relations between the amplitudes
governing $\tau$ decays into four pions and electron positron
annihilation into four pions have been found which allow to
determine all four modes after the amplitude for the $\pi^+\pi^-2\pi^0$
channel  has been fixed. The kinematic breaking of these isospin
relations as a consequence of the $\pi^-$ -- $\pi^0$ mass difference has
also been investigated.

The program is constructed in analogy to
the one \cite{BKM} simulating $e^+e^- \to \gamma  + 2 \pi$. However, it does
not include final state radiation from the charged pions. Additional
collinear photon radiation has been incorporated with the technique of
structure functions. Predictions are presented for cms energies of 1GeV,
3GeV and 10GeV, corresponding to the energies of DAPHNE, BEBC and
of $B$-meson factories.

Even after applying realistic cuts the event rates are sufficiently high
to allow for a precise measurement of $R(Q^2)$ in the region of $Q$
between approximately 1GeV and 2.5GeV.

The model predictions are compared to recent data from electron positron
colliders. Once more accurate data become available, the modular
structure of the program will allow for modification of
replacement of the hadronic current in a simple way.



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\vskip 0.4 cm

{\bf Acknowledgments.}
The work by H.C. was supported by BMBF-POL-239-96, 
the work by J.K. by BMBF under grant BMBF-057KA92P. 
H.C. is grateful for the support and the kind hospitality 
of the Institut f{\"u}r Theoretische Teilchenphysik
 of the Karslruhe University, where this work was carried out. 
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 

\vskip 0.4 cm

\begin{appendix}

\noindent
{\Large \bf Appendix: The description of the hadronic current.}
%\section{Appendix: The description of the hadronic current.}

 In this appendix we give a complete definition of the hadronic current
 used in this paper. Recalling \Eq{sum}
  
\begin{eqnarray}
 \Gamma^{\mu}_{\rho^0 \rightarrow 2\pi^0 \pi^+ \pi^-} =
 \Gamma^{\mu}_{a_1} +\Gamma^{\mu}_{f_0}+\Gamma^{\mu}_{\omega} \ ,
\end{eqnarray}

\noindent
 we will define here its ingredients.

 Denoting  the pions four momenta
 as follows: \(q_1(\pi^0) ,q_2(\pi^0) , q_3(\pi^-) \ \ {\rm and}\ 
 \ q_4(\pi^+) \)
 one
 gets the following contribution from the part containing an \(a_1\) exchange

\begin{eqnarray}
 &&\Gamma^{\mu}_{a_1}\left(q_1,q_2,q_3,q_4\right) = \nonumber \\ 
 &&\tilde \Gamma^{\mu}_{a_1}(q_3,q_2,q_1,q_4) +
 \tilde \Gamma^{\mu}_{a_1}(q_3,q_1,q_2,q_4) -
 \tilde \Gamma^{\mu}_{a_1}(q_4,q_2,q_1,q_3) -
 \tilde \Gamma^{\mu}_{a_1}(q_4,q_1,q_2,q_3). 
 \end{eqnarray}

The structure of this current follows from the form of the Lagrangian
of the \(\rho a \pi \) (\( \sim \vec a \cdot \left(\vec\pi \times 
 \vec \rho\right)\) ) and \(\rho \pi \pi \) 
 (\( \sim \vec \rho \cdot \left(\vec\pi_1 \times 
 \vec \pi_2\right)\) ) interactions and the function 
 \(\tilde \Gamma^{\mu}_{a}\) is defined as \cite{Fink}


\begin{eqnarray}
 \tilde \Gamma^{\mu}_{a_1}(q_1,q_2,q_3,q_4) &=& 
  C_1 \ \
 T_{\rho}\left(Q^2\right)  \ \ \Gamma^{\mu \nu}\left(Q,Q-q_1\right)
 \nonumber \\
  &&BW_{a_1}\left((Q-q_1)^2\right) \ \
 \Gamma_{\nu\lambda }\left(Q-q_1,Q-q_1-q_2\right)\nonumber \\
 &&BW_{\rho}\left((q_3+q_4)^2\right) \ \
 \Gamma_1^{\lambda}(q_3-q_4) \ ,
 \end{eqnarray} 

 where we have followed the notation of \cite{Fink}; 
 \( Q = q_1+q_2+q_3+q_4\).

 Assuming that \(q_1^2 = q_2^2=q_3^2=q_4^2=m_{\pi}^2 \) i.e. that
 pions have equal masses one finds

\begin{eqnarray}
 &&{\kern-30pt}\tilde \Gamma^{\mu}_{a_1}(q_1,q_2,q_3,q_4) = \nonumber \\
  &&C_1 \  f_{\rho\pi\pi}\left((q_3+q_4)^2\right) \ 
          f_{a_1\rho\pi}(Q^2,(Q-q_1)^2) \ f_{a_1\rho\pi}((Q-q_1)^2,(q_3+q_4)^2)
\nonumber \\
 &&T_{\rho}\left(Q^2\right)  \ \ BW_{a_1}\left((Q-q_1)^2\right) \ \
 BW_{\rho}\left((q_3+q_4)^2\right) \nonumber \\
 &&{\kern-7pt} 
 \left[(q_3-q_4)^{\mu} + q_1^{\mu} \frac{q_2(q_3-q_4)}{(Q-q_1)^2}
 -Q^{\mu}\left(\frac{(q_1+q_2)(q_3-q_4)}{Q^2} 
 +\frac{(Q q_1)(q_2(q_3-q_4))}{Q^2 (Q-q_1)^2}\right)\right] \ ,
\end{eqnarray}

where \(f_{\rho\pi\pi}\) and \(f_{a_1\rho\pi}\) describe \(\rho \pi \pi\)
and \(a_1\rho\pi\) vertices appropriately while \(T_{\rho}\),
 \(BW_{a_1}\) and \(BW_{\rho}\) are \(\rho\), \(a_1\) and again \(\rho\)
 propagators. The different choice of the authors of \cite{Fink}
 for two \(\rho\) propagators comes from the fact that \(\rho\)
 couples in a different way in the two cases and the 'propagators' absorb
 in fact some parts of the couplings. The normalization constant 
 \(C_1 =\frac{2 \ \sqrt{6}}{f_{\pi}^2}\) was chosen to give 
 the proper chiral limit of the current. The form factors were chosen
 to be constant 

 \( \left[f_{\rho\pi\pi}\left((q_3+q_4)^2\right) \ 
 f_{a_1\rho\pi}(Q^2,(Q-q_1)^2) \ f_{a_1\rho\pi}((Q-q_1)^2,(q_3+q_4)^2)
 \right]= 0.38\),
 
\noindent
 and the validity of that assumption was proved \cite{Fink} 
 in a limited range of \(Q^2\)  

\noindent
 (\(1.1 \ {\rm GeV} \ < \sqrt{Q^2} \ < \ 2.2 \ {\rm GeV} \) ).
   
 The contribution to the current from \(f_0\) exchange reads

\begin{eqnarray}
 \Gamma^{\mu}_{f_0}(q_1,q_2,q_3,q_4)  &=& \nonumber \\
 &&{\kern-20pt}C_2 \ f_{\rho \rho f_0}\left(Q^2,(q_3+q_4)^2,(q_1+q_2)^2\right) 
 \ f_{\rho\pi\pi}\left((q_3+q_4)^2\right)
                  \ f_{f_0 \pi\pi}\left((q_1+q_2)^2\right)\nonumber \\
 &&T_{\rho}\left(Q^2\right) \ T_{\rho}\left((q_3+q_4)^2\right) \ 
  BW_{f_0}\left((q_1+q_2)^2\right)
\nonumber \\
 &&\left[(q_3-q_4)^{\mu} - Q^{\mu} \frac{Q(q_3-q_4)}{Q^2}\right] \ ,
\end{eqnarray}

 where \(f_{\rho \rho f_0}\) and \( f_{f_0 \pi\pi}\) 
 describe \(\rho \rho f_0 \)
and \( f_0 \pi\pi\) vertices appropriately, while \(BW_{f_0}\) is an 
 \( f_0 \) propagator. 

Again \( C_2 = -\frac{3 \ \sqrt{6}}{f_{\pi}^2}\)
 follows from the chiral limit and the form factors are kept constant
 
\( f_{\rho \rho f_0}\left(Q^2,(q_3+q_4)^2,(q_1+q_2)^2\right) 
 \ f_{\rho\pi\pi}\left((q_3+q_4)^2\right)
                  \ f_{f_0 \pi\pi}\left((q_1+q_2)^2\right) = 0.38 \ . \)

 The contribution coming from so called anomalous (containing \(\omega\)
 exchange) part of the current reads

\begin{eqnarray}
 &&\Gamma^{\mu}_{\omega}(q_1,q_2,q_3,q_4)=
 \frac{g_{\omega}} {\sqrt{2}} \ 1475.98 \ {\rm GeV}^{-3} 
 \ 12.924 \ {\rm GeV}^{-1} \ 0.266 \ m_{\rho}^2 \nonumber \\
 &&{\kern-25pt}
 \left[q_{1}^{\mu} F_1(q_1,q_2,q_3,q_4)+q_{2}^{\mu} F_1(q_2,q_1,q_3,q_4)
  +q_{3}^{\mu} F_2(q_1,q_2,q_3,q_4)-q_{4}^{\mu} F_2(q_1,q_2,q_4,q_3)
  \right] \ ,
\end{eqnarray}

where 
\begin{eqnarray}
&&{\kern-25pt}F_1(q_1,q_2,q_3,q_4)= BW_{\rho,\omega}(Q^2,(Q-q_2)^2)
 \left[(q_3(Q-q_2)) \ (q_2q_4) -(q_4(Q-q_2)) \ (q_2q_3) \right] \\
&&{\rm and}\nonumber \\
&&{\kern-25pt}F_2(q_1,q_2,q_3,q_4)= BW_{\rho,\omega}(Q^2,(Q-q_2)^2)
 \left[(q_4(Q-q_2)) \ (q_1q_2) -(q_1(Q-q_2)) \ (q_2q_4) \right]\nonumber \\
&&{\kern+55pt}+BW_{\rho,\omega}(Q^2,(Q-q_1)^2)
 \left[(q_4(Q-q_1)) \ (q_1q_2) -(q_2(Q-q_1)) \ (q_1q_4) \right] 
\end{eqnarray}

\noindent
and \(g_{\omega} = 1.55\). It is changed from its original value
 \(g_{\omega} = 1.4\) in \cite{Fink} to reproduce the \(\tau\)
 decay rates \(\tau^-\to\nu_{\tau}2\pi^-\pi^+\pi^0\) and 
 \( \tau^-\to\nu_{\tau}\pi^-\omega\left(\pi^-\pi^+\pi^0\right)\).

For completeness we list here all propagators \cite{Fink} required
for the current:

\begin{eqnarray}
T_{\rho}\left(Q^2\right)= \frac{
  BW_3(Q^2,m_{\rho},\Gamma_{\rho}) 
 + \beta_1 BW_3(Q^2,m_{\rho_1},\Gamma_{\rho_1})
 + \beta_2 BW_3(Q^2,m_{\rho_2},\Gamma_{\rho_2})}{1+\beta_1+\beta_2} \ ,
\end{eqnarray}

with

\begin{eqnarray}
 BW_3(Q^2,m_{\rho},\Gamma_{\rho}) = \frac{m_{\rho}^2}
 {m_{\rho}^2-Q^2 - i
 \Gamma_{\rho}m_{\rho}\sqrt{\frac{m_{\rho}^2}{Q^2} 
 \left[\frac{Q^2-4m_{\pi}^2}{m_{\rho}^2-4m_{\pi}^2}\right]^3} }
\end{eqnarray}

and the numerical values set to
\begin{eqnarray}
 && m_{\pi} = 0.14 \ {\rm GeV}\nonumber \\
 && m_{\rho} = 0.773 \ {\rm GeV} \ \ \Gamma_{\rho} = 0.145 
 \ {\rm GeV}\nonumber \\
 && m_{\rho_1} = 1.35 \ {\rm GeV} \ \ \Gamma_{\rho_1} = 0.3 
 \ {\rm GeV} \ \ \beta_1 = 0.08 \nonumber \\
 && m_{\rho_2} = 1.7 \ {\rm GeV} \ \ \Gamma_{\rho_2} = 0.235 
 \ {\rm GeV} \ \ \beta_2 = -0.0075  \ \ ;
\end{eqnarray}

\begin{eqnarray}
BW_{\rho}\left(Q^2\right) = \frac{
  BW_3(Q^2,m_{\rho},\Gamma_{\rho}) 
 + \beta BW_3(Q^2,\tilde m_{\rho_1},\tilde \Gamma_{\rho_1})
  }{1+\beta} \ \ ,
\end{eqnarray}

with 
\begin{eqnarray}
\tilde m_{\rho_1}= 1.37 \ {\rm GeV} \ \ \tilde \Gamma_{\rho_1} = 0.145
 \ {\rm GeV} \ \ \beta = -0.145 \ ;
\end{eqnarray}

\begin{eqnarray}
 BW_{a_1}\left(Q^2\right) = \frac{m_{a_1}^2}
 {m_{a_1}^2-Q^2 - i \Gamma_{a_1}m_{a_1}\frac{g(Q^2)}{g(m_{a_1}^2)}} \ ,
\end{eqnarray}

where
\begin{eqnarray}
 &&g(Q^2) =  1.623 \ Q^2 + 10.38 - \frac{9.32}{Q^2} + \frac{0.65}
        {(Q^2)^2} \ \ \ \ {\rm for}  \ \ \ \ Q^2 > (m_{a_1}+m_{\pi})^2
  \nonumber \\
 &&g(Q^2) =  4.1 \ \left(Q^2-9m_{\pi}^2\right)^3
 \left[ 1 -3.3 \left(Q^2-9m_{\pi}^2\right) + 5.8\left(Q^2-9m_{\pi}^2\right)^2
 \right] \nonumber \\
  &&{\kern+100pt}{\rm for}  \ \ Q^2 < (m_{a_1}+m_{\pi})^2
\end{eqnarray}

and 
\begin{eqnarray}
 m_{a_1} = 1.251 \ {\rm GeV} \ \  \Gamma_{a_1} = 0.599 \ {\rm GeV} \ ;
\end{eqnarray}

\begin{eqnarray}
BW_{f_0}\left(Q^2\right) =  \frac{m_{f_0}^2 - i m_{f_0}\Gamma_{f_0}}
 {m_{f_0}^2 - Q^2 - i m_{f_0}\Gamma_{f_0}}
\end{eqnarray}

with 
\begin{eqnarray}
m_{f_0} = 1.3 \ {\rm GeV} \ \  \Gamma_{f_0} = 0.6 \ {\rm GeV} \ ;
\end{eqnarray}

\begin{eqnarray}
BW_{\rho,\omega}(Q^2,Q_1^2) &=& 
 \left[ \frac{1}{m_{\rho}^2 - Q^2 - i m_{\rho}\Gamma_{\rho}}
 +\sigma \frac{1}{m_{ \rho\prime}^2 - Q^2 - i m_{\rho\prime}
 \Gamma_{\rho\prime}}
 \right] \frac{1}{m_{\omega}^2 - Q_1^2 - i m_{\omega}\Gamma_{\omega}}
 \nonumber \\
 && {\kern-20pt}\left[ \ \theta\left((2.2 \ {\rm GeV})^2 - Q^2\right)
   + \theta\left(Q^2-(2.2 \ {\rm GeV})^2 \right)
 \left(\frac{\left(2.2 \ {\rm GeV}\right)^2}{Q^2}\right)^2 \right]
\end{eqnarray}

with 
\begin{eqnarray}
 m_{ \rho\prime} = 1.7 \ {\rm GeV} \ \  
 \Gamma_{\rho\prime} = 0.26 \ {\rm GeV} \ \ \sigma = -0.1 \nonumber \\
 m_{\omega} = 0.782 \ {\rm GeV} \ \ \Gamma_{\omega}= 0.0085 \ {\rm GeV} \ .
\end{eqnarray}       

 The additional suppression compared to \cite{Fink}  above
 \(Q^2 = \left(2.2 \ {\rm GeV}\right)^2\) was introduced
 "by hand"  
 to get a proper fall off of the cross section at large \(Q^2\).

That completes the definition of the \( \rho^0 \rightarrow 2\pi^0 \pi^+\pi^-\)
 current. One should stress that the assumption that all pion masses are 
 equal in the hadronic current is essential as only then one can expect
 the relations between different components of the \(\rho\) current to hold.
\end{appendix}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%% 
\def\NP{{\sl Nuc. Phys.}} 
\def\PL{{\sl Phys. Lett.}} 
\def\PR{{\sl Phys. Rev.}} 
\def\PRL{{\sl Phys. Rev. Lett.}} 

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\end{document}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%


















